Author's response to reviews

Title: Potent inhibition of rhabdoid tumor cells by combination of flavopiridol and 4OH-Tamoxifen

Authors:

Velasco Cimica (vcimica@notes.cc.sunysb.edu)
Melissa E Smith (melissa.smith@phd.einstein.yu.edu)
Zhikai Zhang (zzhang@aecom.yu.edu)
Deepti Mathur (dm322@cornell.edu)
Sridhar Mani (sridhar.mani@einstein.yu.edu)
Ganjam V Kalpana (ganjam.kalpana@einstein.yu.edu)

Version: 6 Date: 15 November 2010

Author's response to reviews:

Dear Sir or Madam,

We are submitting a final version of the manuscript text and figures. These files include all the changes requested and we have carefully read through the manuscript to eliminate any typographical or other mistakes. Now the manuscript is ready to be published.

Sincerely,

Ganjam V. Kalpana
Professor